Author's response to reviews

Title: Procalcitonin guided antibiotic therapy and hospitalization in patients with lower respiratory tract infections: a prospective, multicenter, randomized controlled trial

Authors:

Philipp Schuetz (schuetzp@uhbs.ch)
Mirjam Christ-Crain (christmj@bluewin.ch)
Marcel Wolbers (MWolbers@uhbs.ch)
Ursula Schild (Schildu@uhbs.ch)
Robert Thomann (thomannr@uhbs.ch)
Claudine Falconnier (falconnierc@uhbs.ch)
Isabelle Widmer (isabelle.widmer@ksl.ch)
Stephanie Neidert (stefanie.neidert@ksl.ch)
Claudine A Blum (claudine.blum@ksa.ch)
Ronald Schoenenberger (rschoenenberger_so@spital.ktso.ch)
Christoph Henzen (christoph.henzen@ksl.ch)
Thomas Bregenzer (thomas.bregenzer@ksa.ch)
Claus Hoess (claus.hoess@stgag.ch)
Martin Krause (martin.krause@stgag.ch)
Heiner C Bucher (Buch erh@uhbs.ch)
Werner Zimmerli (werner.zimmerli@ksli.ch)
Beat Mueller (happy.mueller@unibas.ch)
. for the ProHOSP study group (schuetzp@uhbs.ch)

Version: 3 Date: 5 July 2007

Author's response to reviews:

Dear Editorial Team

Please find attachet our revised manuscript; please contact me anytime for further corrections!

Your abstract must be structured into the following sections: Background, Methods and design, Discussion, Trial registration.

--> ok

Please change Contributors --> Authors' contributions, Acknowledgement --> Acknowledgements, Competing Interest --> Competing interests, REFERENCES --> References and FIGURE AND TABLES LEGENDS --> Figure legends.
Please then add the heading 'Tables' just prior to your Tables.

--> ok

Please remove all underlining in the manuscript text - I noticed it in particular in your figure legends section.
The submission system has only extracted 9 of your 38 references. Please rectify this, making sure that you have mentioned each reference in numerically consecutive order and that you don't group too many references together (e.g. [10-20] might be too large a group for the system).

We used Endnote software, the newest version (Endnote X) for the references; three references were added manually because they are not listed (Ref 15, 21, 23)

Thank you very much for your efforts,

Kind regards

Philipp Schuetz